Author’s response to reviews

Title: Non-collagen genes role in digenic Alport syndrome

Authors:

Sergio Daga (daga@student.unisi.it)
Chiara Fallerini (fallerini2@unisi.it)
Simone Furini (simone.furini@dbm.unisi.it)
Carmine Pecoraro (pecoraro@unina.it)
Francesco Scolari (ceccoscolari@gmail.com)
Francesca Ariani (francesca.ariani@unisi.it)
Mirella Bruttini (mirella.bruttini@unisi.it)
Maria Antonietta Mencarelli (mariaantonietta.mencarelli@unisi.it)
Francesca Mari (francesca.mari@unisi.it)
Alessandra Renieri (alessandra.renieri@unisi.it)
Anna Maria Pinto (annamaria.pinto@dbm.unisi.it)

Version: 3 Date: 14 Feb 2019

Author’s response to reviews:

-- All the email addresses of each author have been included in the title page. The corresponding author has been specified.

-- All authors are individually mentioned in the author contribution statement.

-- The role of the funding body in the design of the study and collection, analysis and interpretation of the data has been specified in the section "Funding"

-- All the figures has been modified as required.

-- The manuscript and the figures has been uploaded as required.